Author's response to reviews

Title: Tolerability of NGX-4010, a capsaicin 8% dermal patch, following pretreatment with lidocaine 2.5%/prilocaine 2.5% cream in patients with post-herpetic neuralgia

Authors:

Lynn R Webster (lrwebstermd@gmail.com)
Margarita Nunez (mnunez@cnsmail.com)
Marvin D Tark (mtark@drugstudies.net)
Edwin D Dunteman (edunteman@aapain.net)
Biao Lu (biao_lu_2002@yahoo.com)
Jeffrey K Tobias (jtobias@neurogesx.com)
Geertrui F Vanhove (tvanhove@depomed.com)

Version: 5 Date: 15 December 2011

Author's response to reviews:

Figure 1 needs to keep the mean NPRS scores at the top of the graph